Author’s response to reviews

Title: An evaluation of Croi MyAction community lifestyle modification programme compared to standard care to reduce progression to diabetes/pre-diabetes in women with prior gestational diabetes mellitus (GDM): study protocol for a randomised controlled trial

Authors:

Jennifer J Infanti (jennifer.infanti@nuigalway.ie)
Fidelma P Dunne (fidelma.dunne@nuigalway.ie)
Angela O’Dea (angela.odea@nuigalway.ie)
Paddy Gillespie (paddy.gillespie@nuigalway.ie)
Irene Gibson (irene.croi@hse.ie)
Liam G Glynn (liam.glynn@nuigalway.ie)
Eoin Noctor (eoinnoctor@physicians.ie)
John Newell (john.newell@nuigalway.ie)
Brian E McGuire (brian.mcguire@nuigalway.ie)

Version: 3 Date: 5 April 2013

Author's response to reviews:

The following revisions have been made to the manuscript:

1. Title changed to conform to journal style for study protocol articles.

2. Abstract re-formatted so it is structured into separate sections headed Background, Methods/Design, Discussion (if appropriate), and the trial registry is listed.

3. Study protocol formatted as outlined in Trials Journal instructions for authors.

4. Ethics statement included in the Methods section of the manuscript (p.7 revised manuscript).

5. Statement included in the Methods section to the effect that informed consent was obtained from each participant (p.9 revised manuscript).